Author's response to reviews

Title: Nrf2 is required to maintain the self-renewal of Glioma stem cells.

Authors:

    Jianhong Zhu Ph. D. (dg1135065@smail.nju.edu.cn)
    Handong Wang M. D.;Ph. D. (hwang_nz@yahoo.cn)
    Qing Sun Ph. D. (zhu86086@gmail.com)

Version: 2 Date: 4 April 2013

Author's response to reviews: see over
Dear Editor:

This is a manuscript by Jianhong Zhu and Handong Wang entitled “Nrf2 is required to maintain the self-renewal of Glioma stem cells”. It is submitted to be considered for publication as a “research article” in your journal. This paper is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal.

We believe the paper may be of particular interest to the readers of your journal as it finding the relationship of Nrf2 and the self-renewal of glioma stem cells. We believe this will be an important target for the treatment of glioblastoma in the future.

Correspondence should be addressed to Handong Wang at the following address: Department of neurosurgery in Jinling Hospital& Neurosurgical Institution of People's Liberation Army of China. No.305, East Zhongshan Road, 210002, Nanjing, Jiangsu, China. Phone and fax number: +86-025-8086-1110, and email address: hdwang_nz@yahoo.cn.

Thanks very much for your attention to our paper!

Sincerely yours,

Jianhong Zhu
Handong Wang